Author's response to reviews

Title: The association between neighborhood greenness and cardiovascular disease: an observational study

Authors:

Gavin Pereira (gavin.pereira@uwa.edu.au)
Sarah Foster (sarah.foster@uwa.edu.au)
Karen Martin (karen.martin@uwa.edu.au)
Hayley Christian (hayley.christian@uwa.edu.au)
Bryan Boruff (bryan.boruff@uwa.edu.au)
Matthew Knuiman (matthew.knuiman@uwa.edu.au)
Billie Giles-Corti (b.giles-corti@unimelb.edu.au)

Version: 2 Date: 24 February 2012

Author's response to reviews: see over
1) Please format the abstract according to our guidelines (http://www.biomedcentral.com/info/ifora/abstracts). The abstracts of manuscripts submitted to the BMC-series should be structured as follows:
- Background: This should place the study into the context of the current knowledge in its field and list the purpose of the work; in other words, the authors should summarise why they carried out their research.
- Methods: This should summarize how the study was performed and mention the different techniques employed. It should also include details of any statistical tests employed.
- Results: This section should describe the main findings of the study.
- Conclusions: A brief summary of the content of the manuscript and the potential implications of its results.

Response:
Amended the objective section to a Background section. Added background, reasons for study and aim. Amended the conclusion to better explain the results and significance of the study.

(2) Please place the Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions.

Response:
Done

(3) Please remove the CFI from the title page and place it as ‘Competing interests’ section in between the ‘Conclusions’ and ‘References’ sections. If there are none to declare, please write 'The authors declare that they have no competing interests'. Please check the instructions for authors on the journal website for a full list of questions to consider when writing your competing interests statement.

Response:
Done. Also rephrased according to the suggested wording.

(4) Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Response:
Done. Included approval body, approval number, and accordance with the Helsinki Declaration.